

GOPEN ACCESS

Citation: Decuypere S, Maltha J, Deborggraeve S, Rattray NJW, Issa G, Bérenger K, et al. (2016) Towards Improving Point-of-Care Diagnosis of Nonmalaria Febrile Illness: A Metabolomics Approach. PLoS Negl Trop Dis 10(3): e0004480. doi:10.1371/ journal.pntd.0004480

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, UNITED STATES

Received: August 13, 2015

Accepted: February 1, 2016

Published: March 4, 2016

Copyright: © 2016 Decuypere et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All metabolite data has been deposited in the metabolomics data repository, MetaboLights (study identifier MTBLS315).

Funding: This work was supported by a Telethon Kids Institute small grant & a Merit Award of the Department of Health Western Australia to SDec, JRC, MM and TR; the Institute of Tropical Medicine Belgium to JM, JJ, GI, KB, PL, MCT and HT; the UK Medical Research Council to NJWR and RG; and the Flemish Ministry of Sciences (EWI, SOFI grant) to SDeb and JJ. The funders had no role in study **RESEARCH ARTICLE**

Towards Improving Point-of-Care Diagnosis of Non-malaria Febrile Illness: A Metabolomics Approach

Saskia Decuypere¹*, Jessica Maltha^{2,3}, Stijn Deborggraeve⁴, Nicholas J. W. Rattray⁵, Guiraud Issa⁶, Kaboré Bérenger⁶, Palpouguini Lompo⁶, Marc C. Tahita⁶, Thusitha Ruspasinghe⁷, Malcolm McConville⁷, Royston Goodacre⁵, Halidou Tinto⁶, Jan Jacobs^{2,8}, Jonathan R. Carapetis¹

 Telethon Kids Institute, University of Western Australia, Perth, Australia, 2 Department of Clinical Sciences, Institute of Tropical Medicine, Antwerp, Belgium, 3 Center for Molecular and Vascular Biology, KU Leuven, Leuven, Belgium, 4 Department of Biomedical Sciences, Institute of Tropical Medicine, Antwerp, Belgium, 5 School of Chemistry, Manchester Institute of Biotechnology, University of Manchester, Manchester, United Kingdom, 6 Clinical Research Unit Nanoro—IRSS-CRUN, Nanoro, Burkina Faso, 7 Metabolomics Australia and Bio21 Institute of Molecular Sciences and Biotechnology, University of Melbourne, Melbourne, Australia, 8 Department of Immunology and Microbiology, KU Leuven, Leuven, Belgium

* saskia.decuypere@telethonkids.org.au

Abstract

Introduction

Non-malaria febrile illnesses such as bacterial bloodstream infections (BSI) are a leading cause of disease and mortality in the tropics. However, there are no reliable, simple diagnostic tests for identifying BSI or other severe non-malaria febrile illnesses. We hypothesized that different infectious agents responsible for severe febrile illness would impact on the host metabololome in different ways, and investigated the potential of plasma metabolites for diagnosis of non-malaria febrile illness.

Methodology

We conducted a comprehensive mass-spectrometry based metabolomics analysis of the plasma of 61 children with severe febrile illness from a malaria-endemic rural African setting. Metabolite features characteristic for non-malaria febrile illness, BSI, severe anemia and poor clinical outcome were identified by receiver operating curve analysis.

Principal Findings

The plasma metabolome profile of malaria and non-malaria patients revealed fundamental differences in host response, including a differential activation of the hypothalamic-pituitary-adrenal axis. A simple corticosteroid signature was a good classifier of severe malaria and non-malaria febrile patients (AUC 0.82, 95% CI: 0.70–0.93). Patients with BSI were characterized by upregulated plasma bile metabolites; a signature of two bile metabolites was estimated to have a sensitivity of 98.1% (95% CI: 80.2–100) and a specificity of 82.9% (95% CI:

design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

54.7–99.9) to detect BSI in children younger than 5 years. This BSI signature demonstrates that host metabolites can have a superior diagnostic sensitivity compared to pathogendetecting tests to identify infections characterized by low pathogen load such as BSI.

Conclusions

This study demonstrates the potential use of plasma metabolites to identify causality in children with severe febrile illness in malaria-endemic settings.

Author Summary

In the tropics, malaria is commonly attributed to be the cause of most childhood fevers, while in fact this condition is more commonly caused by other pathogens that are clinically indistinguishable from malaria. These so-called non-malaria febrile illnesses include bacterial bloodstream infections, which are associated with a higher mortality than malaria. Most health care facilities in the tropics have malaria diagnostic tests available, but tests for non-malarial febrile illnesses are extremely limited. There is the critical need for new tests that can address the question 'if a febrile patient is not suffering from malaria, then what is it and what treatment will be effective?' Using metabolomics, we have comprehensively screened the biochemical profile of patients with severe febrile illness for biological markers of non-malaria febrile illness. The results show that severe malaria and non-malaria febrile illness trigger a distinct metabolic response in the host. We demonstrate that this pathophysiological difference can be exploited for differential diagnosis of severe febrile illness and identification of patients with bacterial bloodstream infections.

Introduction

The introduction of malaria rapid diagnostic tests (RDT) has revealed that febrile illnesses in the tropics and subtropics are more commonly caused by non-malaria pathogens than by malaria [1-5]. Bacterial bloodstream infections (BSI) are considered the most severe non-malaria febrile illness, with mortality rates of 10-25% [6-8]. The high BSI mortality rates high-light the importance of accurate diagnosis and immediate correct case management, particularly in malaria-endemic regions where clinical presentation of BSI and severe malaria are similar [9].

However, in most malaria-endemic regions, children with non-malaria severe febrile illness are not easily diagnosed, which reflects a number of deficiencies in current WHO guidelines that still rely largely on malaria diagnostic tests. First of all, health care workers do not have access to tests to diagnose non-malaria febrile illness, which would inform the most effective treatment. For instance, BSI diagnosis still relies on traditional microbiology, which requires laboratory infrastructure and highly trained staff. Even when available, it takes two to three days for a result, which is too slow to timely inform case management. This can lead to ineffective first-line treatment choices that result in poorer survival outcomes [10], or alternatively leads to over-treatment with broad-spectrum antibiotic therapy that wastes limited resources and fuels emerging antibiotic resistance [11]. Secondly, the current guidelines also fail to alert for primary non-malaria febrile illness in patients with asymptomatic or recent malaria that can have a positive malaria RDT [12]. Finally, the current guidelines fail to recognize concomitant non-malaria febrile illness in patients with severe malaria. BSI/malaria co-infections affect

5–7% of all children with malaria in Africa, and have a 2–3 fold higher mortality rate compared to patients with malaria alone [6].

An ideal test to assess severe febrile illness in malaria-endemic settings should detect more than a single pathogen. In particular, it should provide information on the type of infecting pathogen (malaria and non-malaria) in order to immediately inform on the best treatment and referral options. Such information is unlikely to be captured by a single biological measurement or molecule (biomarker), but is expected to require a combination of molecules (signature). With the development of 'omics' profiling approaches it is now possible to measure hundreds of biological analytes simultaneously and thus enable assessment of the diagnostic performance of a molecular signature. This study utilized metabolomics, which in contrast to studies of DNA, RNA and proteins, enables characterization of the metabolome that is the final product of all cell regulatory processes. Hence, a metabolome profile provides a read-out of the (patho) physiological status of a patient at the time of sampling that cannot be obtained directly from the genome, transcriptome, or proteome. In addition, metabolomics analyses only require small sample volumes $(20-50 \ \mu L)$ to determine a comprehensive metabolome profile, which is a particular advantage when studying pediatric populations. We hypothesized that the pathophysiological processes triggered by malaria and non-malaria febrile illness induce distinctive changes in the > 4,000 blood metabolites and explored for the first time whether such characteristic metabolites can be used for differential diagnosis of severe febrile illness.

Materials and Methods

Ethics statement

The study was conducted according to the principles expressed in the Declaration of Helsinki and was approved by the national ethics committee of Burkina Faso, the institutional review board of the Institute of Tropical Medicine Antwerp, the ethics committee of the University Hospital of Antwerp and the human research ethics office of the University of Western Australia. Written informed consent was given by all parents or guardians of enrolled children.

Study site, patients and sample collection

Patients were recruited at the district hospital Centre Médical avec Antenne Chirurgicale Saint Camille in malaria-endemic Nanoro, Burkina Faso. Admitted children < 15 years of age presenting with axillary temperature $\geq 38^{\circ}$ C, or clinical signs of severe illness were enrolled. Medical history, physical examination and outcome of febrile episode were registered on a standardized form. At time of hospital admission, venous whole blood for blood culture, malaria diagnosis, full blood count, glucose measurements, plasma metabolome analysis, and 16S rRNA deep sequencing were collected from all participants by trained study nurses. Details of sample processing and diagnostic procedures are provided in <u>S1 Methods</u>.

Case definitions

Malaria was defined as the presence of asexual *Plasmodium falciparum* parasites in blood smear confirmed by microscopy. All patients with a negative blood smear were classified as non-malaria. Recent malaria was defined as positive malaria HRP-2 RDT (which can remain positive up to 6 weeks after successful treatment), but a negative blood smear [13,14]. Confirmed BSI was defined as the growth of clinical significant organisms from blood culture and/ or reproducible detection of clinical significant organisms in two 16S deep sequencing experiments. Patients for which 16S deep sequencing could not be performed (n = 5) were classified as 'BSI diagnosis incomplete'; and patients with unusual but possible clinically relevant bacteria

(n = 4) were classified as 'possible BSI'. Severe anemia was defined as patients with hemoglobin levels < 5 g/dL. Patients who died in hospital following admission for severe febrile illness were classified as non-survival.

Metabolomics analysis

Mass spectrometry-based metabolomics data were collected for all patients with 3 complementary analytical platforms: gas chromatography mass spectrometry (GC-MS), C_8 column liquid chromatography mass spectrometry (C_8 -LC-MS) in positive ionization mode, and C_{18} column ultra-high pressure liquid chromatography mass spectrometry (C_{18} -UHPLC-MS) in positive and negative ionization mode. These three analytical techniques each capture a different part of the plasma metabolome due to the different chromatography column chemistries used, and thus provide a very broad coverage of the metabolome. Detailed procedures for sample preparation and data acquisition are provided in <u>S1 Methods</u>. Raw data were processed with dedicated data processing pipelines, which are described in <u>S1 Methods</u>. All metabolite data has been deposited in the metabolomics data repository, MetaboLights (study identifier MTBLS315).

Statistical analysis

All data processing and statistical analyses were performed using the R software environment (version 3.1). This language comprises a selection of packages suitable for the implemented statistical methods: multivariate analysis (partial-least-squares (PLS) regression), receiver-operating curve (ROC) analysis, correlation analysis, hierarchical cluster analysis, biomarker validation and Bayesian latent class modeling. The specific statistical methods and R packages used are explained in <u>S1 Methods</u>, the R scripts can be obtained upon request from the authors.

Results

We studied the plasma metabolome profile in 61 children with severe febrile illness. All were tested for BSI and malaria. The final diagnosis and characteristics of the patients are shown in <u>Table 1</u>. More than half of the cases in the non-malaria group were diagnosed with community-acquired BSI. The BSI organisms included both Gram-negative and Gram-positive species (<u>Table 2</u>). Non-malaria patients with negative diagnosis for BSI and malaria could be BSI patients that were missed by the used BSI diagnostics that are known to have a limited sensitivity, or could be patients infected with other parasites or viruses that were not standardly tested in the study setting.

The plasma metabolome in patients with severe febrile illness

A total of 2,635 reproducibly detected plasma *metabolite features* showed a minimal degree of variation (relative standard deviation < 15%) in the study sample (Table 3), these features were further analyzed to fathom the metabolic nature of severe febrile illness. We first investigated with partial-least-squares regression analyses which characteristics of our study participants have a considerable influence on the plasma metabolome composition (Fig 1). Of all the tested patient characteristics, the correlated factors age/weight/height had the biggest impact on the measured metabolome (Q2 = 61.5/53.1/69.3%), closely followed by blood glucose level (Q2 = 48.1%) and disease outcome (survival Q2 = 46.5%). In comparison, the type of infection (BSI, malaria) had little impact on the measured metabolome (Q2 = 18.4% and 23.1%

	malaria febrile illness	non-mala	aria febrile illness	malaria co-infection			
	malaria	BSI	negative for BSI & malaria	positive for BSI & malaria			
no. of patients	30	14	13	4			
% female	50%	50%	30.8%	50%			
mean (range) age (months)	39.5 (6–147)	39.5 (0–170)	41 (2–145)	26.6 (2–60)			
mean (range) duration illness prior to admission (dys)	4.0 (1–7)	4.3 (1–15)	4.4 (1–6)	5 (3–30)			
% reported prior treatment *	33.3%	64.3%	61.5%	0%			
mean (range) parasite density (x10 ³ par./mL)	20.8 (0.1–204.2)	0	0	15.1 (2.6–109.4)			
mean (range) Hb (g/dL)	7.5 (2.2–12.8)	7.5 (3.8–13.5)	7.5 (3.2–15.9)	6.7 (3.8–8.5)			
mean (range) differential neutrophils (% of total WBC)	51.7 (15.0–90.4)	50.9 (15.5– 88.4)	49.1 (19.4–79.6)	43.0 (10.0–55.2)			
% severe malnutrition **	10%	7%	0	75%			
% non-survival	10%	50%	38.5%	0%			

* Reported prior treatment was defined as reported intake of antimalarial and/or antibiotic treatment in the 48 hrs prior to hospital admission.

** Severe malnutrition was defined as weight for height score < 70% according to national guidelines or report of kwashiorkor.

doi:10.1371/journal.pntd.0004480.t001

respectively), which does however not preclude the existence of individual metabolite features that are characteristic for the type of infection.

Metabolic characteristics of severe febrile illness

We identified the metabolite features characterizing the following five patient groups with ROC analysis: (i) patients with non-malaria febrile illness, (ii) malaria patients, (iii) BSI patients, (iv) non-survival patients and (v) severe anemia patients (<u>Table 4</u>). The details of the metabolite features are provided in <u>S1 Data</u>. This section focuses on the metabolite characteristics of non-malaria illness, malaria and BSI; the results for non-survival and severe anemia are described in <u>S1 Results</u>.

We identified a group of 10 correlating metabolite features that characteristically appeared in non-malaria patients (sensitivity range: 0.90–0.67; specificity range: 0.89–0.63; Fig 2). These features include four corticosteroids, of which three were putatively identified as

Table 2. Non-malarial clinical	ly significant organisms	identified in the study participants.
--------------------------------	--------------------------	---------------------------------------

Clinically significant organisms	No. (%)
all pathogenic bacteria	27 (100%)
Salmonella enterica	9 (33.3%)
Escherichia coli	4 (14.1%)
Streptococcus pneumoniae	3 (11.1%)
Shigella spp.	3 (11.1%)
Enterobacter cloacae	2 (7.4%)
Neisseria meningitidis	2 (7.4%)
Haemophilus influenzae	1 (3.7%)
Klebsiella spp.	1 (3.7%)
Staphylococcus aureus	1 (3.7%)
Pantoea agglomerans	1 (3.7%

doi:10.1371/journal.pntd.0004480.t002

Mass-spectrometry technology	Coverage plasma of the metabolome	No. of reproducible features with RSD < 15%	No. of features putatively identified	No. of features characterizing severe febrile illness
GC—MS	polar metabolites	90	50 (55.6%)	2
C ₈ LC—MS	lipids	548	378 (69.0%)	25
C18 UHPLC-MS	neutral and lipid metabolites	1,997	1,562 (78.2%)	54

Table 3. Overview plasma metabolome coverage.

Abbreviations: GC-MS = gas chromatography mass spectrometry, $C_8 LC-MS = C_8$ column liquid chromatography mass spectrometry, $C_{18} UHPLC-MS = C_{18}$ column ultra-high pressure liquid chromatography mass spectrometry, RSD = relative standard deviation.

doi:10.1371/journal.pntd.0004480.t003

glucocorticoids. The correlation map shows that these corticosteroids were also markers of non-survival patients (S1 Results). In addition, the non-malaria features included three highly correlating features of which one was putatively identified as an eicosanoid (leukotriene F4). Malaria patients on the other hand were characterized by a higher concentration of 16 lipids, predominantly triglycerides and ether phospholipids (sensitivity range: 0.83–0.60; specificity: 0.96–0.63). The latter group included four lipids that showed a fairly positive quantitative correlation with *Plasmodium* parasite density (Pearson's correlation: 0.38–0.45). The heatmap of the 26 metabolite features characteristic for non-malaria/malaria further clarifies the distinct metabolic character of the two patient groups (Fig 3). Non-malaria patients formed two subclusters. Subcluster A had moderate increases of corticosteroids and predominantly consisted of patients who survived the febrile illness episode, while subcluster B had the highest concentration of corticosteroids seen in our study, and was associated with non-survival. The malaria patients also grouped in two major subclusters. Subcluster C was characterized by more moderate increases of the typical malaria lipids compared to subcluster D which may reflect a differential response that varies with age and/or Hb-levels (higher median age and Hb levels in subcluster C compared to subcluster D). Finally, we expected patients with concomitant malaria and non-malaria febrile illness to have an increase in both the typical malaria lipids/triglycerides and the non-malaria metabolites. Such a cluster of 8 presumptive co-infection patients was indeed observed (far right within subcluster D), of whom two were confirmed as a BSI/malaria co-infection and one as a recent malaria case with a BSI infection.

BSI patients were found to have eight upregulated lipids, including several bile acids and alcohols, compared to patients without BSI (sensitivity range: 0.83–0.67; specificity range: 0.91–0.73; <u>Fig 4</u>). We could not find a plausible identification in the checked databases for half of the upregulated features, however given their correlating signal intensity profiles and chromatographic retention-time they were likely biologically and structurally related to the identified bile acids and alcohols. The plasma concentration of these upregulated bile features did not seem to be related to the particular infecting bacterial species.

The metabolome features that had lower concentrations in BSI patients compared to non-BSI patients were consistent with a lower rate of lipolysis (*e.g.* lower concentration of lysophospholipids) and a disruption of fatty acid ß-oxidation (*e.g.* lower concentration of plasma acylcarnitines).

Diagnostic signatures for differential diagnosis of severe febrile illness

We selected a signature of two metabolites for non-malaria patients, and also for BSI patients. We assessed the diagnostic performance of the sum of the signal intensity of the 2 individual metabolites for non-malaria febrile illness and BSI respectively (Fig 5 and Fig 6).



Fig 1. Spiderplot showing the results of the 27 regression analyses modeling the co-variance between metabolome measurements and the listed patient characteristics. The plotted validation metric Q2 allows assessing the relative impact of the patient characteristics on the metabolome. The following 27 patient variables were included in the analysis from center top in clockwise direction: BSI = confirmed or no confirmed BSI diagnosis; malaria RDT = positive or negative malaria rapid diagnostic test; parasite density = number of asexual *Plasmodium* parasites in the bloodstream (x10³/mL); prior AB Rx = reported or no reported antibiotic treatment in the past 48 hrs; prior AM Rx = reported or no reported antimalarial treatment in the past 48 hrs; temp. = axillary temperature at time of hospital admission (°C); duration illness = duration illness prior to hospital admission (days); survival = clinical outcome survival or no survival; sex = male or female; age = age expressed in months; weight = weight expressed in kg; height = height expressed in cm; malnutrition = presence or absence of severe malnutrition defined as weight for height score < 70% according to national guidelines or report of kwashiorkor; blood glucose = blood glucose level (mg/dL); WBC = white blood cell count in blood (x10³/µL); RBC = red blood cell count in blood (x10⁶/µl); hemoglobin = blood hemoglobin level (g/dL); hematocrit = blood hematocrit (%); MCV = mean corpuscular volume (fL); MCH = mean corpuscular hemoglobin (pg); MCHC = mean corpuscular hemoglobin concentration (g/dl); platelet count = platelet blood count (x10³/µL); % neutr. = neutrophils as % of total WBC; % lymph. = lymphocytes as % of total WBC; % monoc. = monocytes as % of total WBC; % eosin. = eosinophils as % of total WBC; % basoph. = basophils as % of total WBC.

doi:10.1371/journal.pntd.0004480.g001

The metabolite signature for non-malaria illness consisted of a C21-corticosteroid (m/z M-H = 377.196, putative ID: $C_{21}H_{30}O_6$ 18-hydroxycortisol, ID_2094_ON in S1 Data) and a steroid glucuronide (m/z M-H = 481.243, putative ID: $C_{25}H_{38}O_9$ 11-beta-hydroxyandrosterone-3-glucuronide, ID_3388_ON in S1 Data). The first steroid had an individual area under the curve or AUC = 0.81 (95% CI = 0.68–0.91), adjusted p-value = 0.01, fold change = 4.90; while the second steroid had an AUC = 0.79 (95% CI = 0.67–0.90), adjusted p-value = 0.01 and fold change = 24.6. The ROC curve analysis for the 2 steroids combined including 57 patients (excluding confirmed malaria/BSI co-infections) had a more favorable AUC value of 0.82 (Fig 5A). As shown in the boxplot of Fig 5A, when considering all 61 patients, there were 8 malaria

Phenotype			upregula	ated features		downregulated features						
	no. features	mean AUC	range fold change	range adjusted p- values	main pathways	no. features	mean AUC	range fold change	range adjusted p- values	main pathways		
non-malaria27 cases vs 30 controls	10	0.77	1.5–34.2	0.01–0.05	Sterol lipids (corticosteroids)	16	0.77	1.5–12.0	0.01–0.06	Triglycerides & phospholipids		
BSI18 cases vs 33 controls	8	0.812	1.5–2.9	0.01–0.1	Sterol lipids (bile acids/alcohols)	8	0.79	1.5–3.3	0.07–0.1	Fatty acid oxidation		
non- survival15 cases vs 45 controls	56 (18)*	0.81	2.4–39.3	0.02–0.05	Sterol lipids (corticosteroids) & amino acids	64 (7)*	0.81	2.01–2.6	0.02–0.05	Triglycerides & phospholipids		
severe anemia17 cases vs 44 controls	67 (11) **	0.81	1.5–20.7	0.04–0.05	Triglycerides	10 (3)**	0.81	1.6–1.8	0.04–0.05	Fatty acids & sphingolipids		

Table 4. Summary metabolite features characteristic for patients with non-malaria, BSI, non-survival and severe anemia.

For each analysis, we selected all features with an AUC > 0.75 (excellent to fair classifiers), an adjusted p-value < 0.1 and a fold change in median signal intensity between the cases and controls > 1.5. For the non-malaria analysis, patients with BSI/malaria co-infection were excluded. For the BSI analysis, patients with BSI/malaria co-infection and patients with incomplete or possible BSI diagnosis were excluded. For the non-survival analysis, patients who left hospital against medical advice were excluded. The metabolic features of non-survival and severe anemia patients are further detailed in <u>S1 Results</u>. * Feature characteristics only shown for the number of features shown between brackets which is the subset of features that have an AUC > 0.75, fold change median signal intensity between 2 compared groups > 2 and p-value < 0.05.

** Feature characteristics only shown for the number of features shown between brackets which is the subset of features that have an AUC > 0.75, fold change median signal intensity between 2 compared groups > 1.5 and p-value < 0.05.

doi:10.1371/journal.pntd.0004480.t004

patients falsely classified as non-malaria by this signature, which included six patients that were presumptive malaria/non-malaria co-infections as indicated in subcluster D of Fig 3. The non-malaria signature failed to identify six non-malaria patients, including two BSI patients with grown blood culture.

As explained above, when present at high concentrations the same corticosteroid metabolites were predictive of non-survival. This is confirmed by the ROC-curve analysis on 60 patients (excluding the single patient who left hospital against medical advice) which had a very good AUC of 0.86, and a cut-off value that was indeed almost two-fold higher than the cut-off value to predict non-malaria febrile illness (Fig 5B). A test based on the two combined metabolites missed only 2/15 non-survival patients and thus had a good sensitivity (87%) to predict non-survival. However the specificity (73%) was rather poor, in our patient sample 12/ 45 survival patients tested false positive (Fig 5B).

We selected a bile acid (m/z M+FA-H = 493.316, putative ID: $C_{27}H_{44}O_5$ C27 bile acid, ID_6663_ON in S1 data) and bile alcohol (m/z M+FA-H = 483.331, putative ID: $C_{26}H_{46}O_5$ 27-Nor-5b-cholestane-3a,7a,12a,24,25-pentol bile alcohol, ID_4741_ON in S1 data) for the BSI signature. The individual diagnostic performance of the bile alcohol was AUC = 0.82 (95% CI = 0.69–0.93), adjusted p-value = 0.07, fold change = 2.54; and for the bile acid AUC = 0.79 (95% CI = 0.63–0.92), adjusted p-value = 0.1, fold change = 2.8. The ROC-curve analysis for the 2 bile metabolites combined including 51 patients (excluding confirmed malaria/BSI coinfections, patients with incomplete or possible BSI diagnosis) had a good AUC of 0.85 (Fig 6). The signature falsely classified 4/18 confirmed BSI cases as non-BSI (Fig 6). Notably these false positives represented 4/5 BSI patients older than 5 years, suggesting that the signature was most suitable to diagnose BSI in under-fives. Eight patients (all malaria patients) in whom we



Fig 2. Plot showing correlation between clinical patient data and the metabolite features characterizing non-malaria illness. The plot consists of 2 panels: (i) in the lower panel, the color and size of the circles correspond to the strength of the correlation, with increasing circle size and color intensity indicating increasing correlation; shades of blue are used for negative correlations and shades of red for positive correlations, crosses indicate correlations that were statistically insignificant (p-value < 0.05), and (ii) the upper panel shows the corresponding Pearson's correlation coefficient r. The left labels for metabolites include the feature ID of the metabolite as listed in S1 data and one putative identification. The following eight patient variables were included: confirmed BSI = confirmed or no confirmed BSI diagnosis; malaria = positive or negative malaria blood smear; parasite density = number of asexual *Plasmodium* parasites in the bloodstream (x10³/mL); disease duration = duration illness prior to hospital admission expressed in days; non-survival = clinical outcome survival; age = age expressed in months; Hb = blood hemoglobin level (g/dL); % neutrophils = neutrophils as % of total WBC. Lipids are abbreviated with PC, PE, SL, DG or TG for phosphatidylcholines, phosphatidylethanolamines, sphingolipids, diacylglycerides and triacylglycerides respectively, and the total number of acyl side-chain carbons and the double bonds in the side-chains.

doi:10.1371/journal.pntd.0004480.g002

could not detect BSI were classified by the metabolite signature as BSI (Fig 6). It is difficult to conclude whether they were falsely classified by the metabolite signature as BSI-positive, or whether they were misdiagnosed by blood culture/sequencing. We therefore estimated sensitivities and specificities of the three BSI diagnostic tests with Bayesian latent class models (Table 5). Although the confidence intervals were fairly wide given the limited sample size, the results were in line with the expectations. Regardless which age group we considered, the



Fig 3. Heatmap of the top 26 features that characterize non-malaria and malaria febrile illness. Patients are shown along the y-axis, the four colorcoded bars right of the heatmap indicate their malaria status, survival outcome, age and severe anemia status. The annotation right of the heatmap shows which pathogenic bacteria were identified by blood culture (BC) and/or by 16S sequencing (16S) along with the pathogen species detected (in case of 16S results, only the most abundant pathogen is shown with the matching number of sequencing reads); NA = incomplete BSI diagnosis. The 26 metabolites are shown along the x-axis, their putative identity and feature ID as listed in S1 data are shown below the heatmap. Unsupervised hierarchical clustering of the patients (the tree right of the y-axis) reveals that the shown metabolite intensity profiles differ sufficiently to distinguish non-malaria (upper branch) and malaria patients (lower branch). Clustering of the metabolites according to similarity in intensity profiles (the tree above the x-axis) reveals 3 major groups of metabolite features: the right cluster are plasma phospholipids and fatty acids typically appearing in malaria patients, the middle cluster are the malaria triglycerides, and the left cluster are metabolites characteristic for non-malaria febrile illness. (Abbreviations: LAMA = left against medical advice, mo. = months, Hb = hemoglobin, lipids are abbreviated with PC, PE, SL, DG or TG for phosphatidylcholines, phosphatidylethanolamines, sphingolipids, diacylglycerides and triacylglycerides respectively, and the total number of acyl side-chain carbons and the double bonds in the side-chains.)

doi:10.1371/journal.pntd.0004480.g003

sensitivity of blood culture (50%) was inferior to 16S sequencing (70%). The BSI metabolite signature had the best sensitivity and negative predictive value of the three tests, and reached an impressive 98.1% and 98.3% respectively when considering under-fives only. However, the specificity of the BSI signature (85.8% in all patients, 82.9% in under-fives only) was inferior to that of blood culture and 16S sequencing (> 95% for both tests).

Discussion

We report here for the first time that malaria and non-malaria severe febrile illnesses each trigger a distinct metabolic host response affecting plasma lipid profiles and this opens new options for differential diagnosis of severe febrile illness. For BSI, one of the most severe nonmalaria febrile illnesses, we identified a simple bile metabolite signature with a superior sensitivity and negative predictive value than the current tests (blood culture and 16S sequencing). If our findings can be validated in large-scale studies, then such a simple metabolite signature could be the basis of a new rapid diagnostic test that could potentially reduce BSI mortality by facilitating early diagnosis and timely hospital referral of BSI patients, and reduce empirical antibiotic usage.

Severe malaria was shown to affect the plasma lipid profile, with triglycerides and phospholipids being most significantly changed (Fig 3). Hypertriglyceridemia was also found in all patients with severe anemia (Table 4, S1 Results). A meta-study on the impact of malaria on plasma lipids supports our findings and demonstrated that malaria is characterized by (i) low serum total cholesterol, HDL and LDL which seems to be unique for malaria, and (ii) high triglycerides which is common to other febrile conditions [15]. The underlying biological mechanism of lipid profile changes during malaria is not fully understood yet [15], but may be related to metabolic changes induced by both the parasite and host immune responses. Our data further suggests that some malaria phospholipid markers are moderately quantitatively correlated with parasite density, and that the overall concentration of the typical malaria lipids increases with disease severity (young age and low Hb-levels, Fig 3). Similar findings have been reported by malaria metabolomics and clinical observational studies [16,17], and suggest that host metabolites characteristic for malaria may be more suitable to assess malaria disease progression than to diagnose malaria patients amongst febrile patients.

In contrast to most malaria patients, non-malaria patients were characterized by the presence of immunoregulatory metabolites including corticosteroids, of which the majority were glucocorticoids, and presumably several metabolites related to the eicosanoids. Glucocorticoids are produced by the adrenal glands in response to activation of the hypothalamic-pituitaryadrenal (HPA) axis by the immune system when sensing challenges like infectious agents [18]. During infection these steroid hormones have immuno-suppressive effects to prevent overshooting of inflammatory and immune responses that would be detrimental [19]. Numerous studies have documented HPA activation and subsequent glucocorticoid release upon bacterial

	ID_5730_ON	ID_6663_ON	ID_177_GC	ID_192_GC	ID_2438_OP	ID_4741_ON	ID_4494_ON	disease duration	ID_17765_OP	confirmed bBSI	non-survival	ЧЧ	Age	% neutrophils	parasite density	malaria	ID_3193_ON	ID_4464_LIP	ID_12445_LIP	ID_4713_LIP	ID_1820_OP	ID_2243_0P	ID_21164_LIP	ID_24913_ON
ID_5730_ON / C26H46O6 C26-bile alcohol		0.73	0.63	0.59	0.65	0.71	0.53	0.23	0.3	0.46	0.17	-0.41	-0.33	-0.26	-0.01	-0.14	-0.54	-0.56	-0.35	-0.42	-0.56	-0.57	-0.51	-0.56
ID_6663_ON / C27H44O5 C27-bile acid			0.48	0.42	0.42	0.68	0.41	0.25	0.33	0.46	0.15	-0.34	-0.47	-0.35	-0.06	-0.22	-0.38	-0.52	-0.31	-0.27	-0.48	-0.48	-0.38	-0.47
ID_177_GC / No ID				0.86	0.66	0.51	0.33	0.33	0.37	0.46	0.01	-0.37	-0.28	-0.07	-0.09	-0.29	-0.45	-0.3	-0.1	-0.25	-0.34	-0.45	-0.31	-0.29
ID_192_GC / No ID			Ŏ		0.69	0.6	0.36	0.27	0.25	0.4	-0.16	-0.44	-0.24	-0.15	0.01	-0.2	-0.34	-0.16	0.01	-0.14	-0.28	-0.4	-0.33	-0.2
ID_2438_OP / No ID				Ŏ		0.69	0.19	0.14	0.27	0.41	-0.09	-0.42	-0.27	-0.14	0	-0.28	-0.49	-0.37	-0.15	-0.31	-0.33	-0.38	-0.45	-0.39
ID_4741_ON / C26H46O5 C26-bile alcohol							0.15	0.19	0.28	0.48	-0.07	-0.34	-0.18	-0.13	0.04	-0.09	-0.48	-0.43	-0.34	-0.39	-0.42	-0.51	-0.51	-0.43
ID_4494_ON / No ID								0.18	0.24	0.36	-0.01	-0.26	-0.26	-0.18	-0.18	-0.11	-0.22	-0.34	-0.22	-0.22	-0.34	-0.32	-0.32	-0.39
disease duration							•		0.46	0.4	-0.02	-0.03	0.14	-0.01	-0.19	-0.15	-0.15	-0.26	-0.11	-0.27	-0.2	-0.21	-0.12	-0.05
ID_17765_OP / C39H76NO8P PE(34:1)										0.53	0.1	0.04	-0.03	-0.02	-0.09	-0.07	-0.53	-0.43	-0.45	-0.57	-0.43	-0.37	-0.45	-0.51
confirmed bBSI											0.21	0.05	0.06	0.14	-0.18	-0.44	-0.48	-0.47	-0.42	-0.48	-0.41	-0.4	-0.52	-0.45
non-survival	×	×	\times	×	\times	\times	\times	Х		•		0.25	0.06	-0.08	-0.08	-0.41	-0.08	-0.35	-0.23	-0.28	-0.06	0.03	-0.26	-0.34
Hb								Х	\times	\times	×		0.46	0.39	-0.21	-0.11	0.06	0	-0.24	-0.18	0.15	0.31	0.21	0.16
Age								×	Х	×	Х			0.62	-0.17	-0.02	0.19	0.23	0	-0.1	0.18	0.24	0.2	0.24
% neutrophils			\times			\times		Х	\times	×	\times				-0.21	-0.17	0.01	0.03	-0.21	-0.22	0.16	0.2	0.24	0.09
parasite density	\times	\times	\times	\times	\times	\times	\times		\times	\times	\times	\times	×			0.46	0.06	0.19	0.17	0.11	-0.1	0.01	-0.06	-0.08
malaria												\times	\times	\times			0.19	0.37	0.18	0.17	-0.05	0.02	0.09	0.07
ID_3193_ON / C46H84NO7P PC(O-38:5)												\times		\times	\times	•		0.7	0.67	0.65	0.51	0.58	0.5	0.58
ID_4464_LIP / C48H86NO7P PC(O-40:6)												\times		\times	×				0.75	0.77	0.45	0.5	0.54	0.66
ID_12445_LIP / C26H56NO6P PC(O-18:0)												\times	\times	×	×					0.86	0.36	0.43	0.44	0.62
ID_4713_LIP / C24H52NO6P PC(O-16:0)												\times	\times	×	×						0.4	0.44	0.59	0.67
1820_OP / C21H37NO4 tetradecadiencarnitine											×			×	\times	\times						0.73	0.6	0.55
ID_2243_OP / C25H45NO4 Linoleyl carnitine											Х			×	X	\times							0.49	0.49
ID_21164_LIP / C31H58O5 DG(14:1/14:0/0:0)														×	Х									0.7
ID_24913_ON / C33H66NO8P PE(28:0)												×		\times	\times									

Fig 4. Plot showing correlation between clinical patient data and the metabolite features characterizing BSI. The plot consists of 2 panels: (i) in the lower panel, the color and size of the circles correspond to the strength of the correlation, with increasing circle size and color intensity indicating increasing correlation; shades of blue are used for negative correlations and shades of red for positive correlations, crosses indicate correlations that were statistically insignificant (p-value < 0.05), and (ii) the upper panel shows the corresponding Pearson's correlation coefficient r. The left labels for metabolites include the feature ID of the metabolite as listed in S1 data and one putative identification. The following eight patient variables were included: confirmed BSI = confirmed or no confirmed BSI diagnosis; malaria = positive or negative malaria blood smear; parasite density = number of asexual *Plasmodium* parasites in the bloodstream (x10³/mL); disease duration = duration illness prior to hospital admission expressed in days; non-survival = clinical outcome survival or no survival; age = age expressed in months; Hb = blood hemoglobin level (g/dL); % neutrophils = neutrophils as % of total WBC. Lipids are abbreviated with PC, PE, SL, DG or TG for phosphatidylcholines, phosphatidylethanolamines, sphingolipids, diacylglycerides and triacylglycerides respectively, and the total number of acyl side-chain carbons and the double bonds in the side-chains.

doi:10.1371/journal.pntd.0004480.g004

ID_

and viral infection [18], but little is known about the HPA axis response during malaria [20– 24]. One study also reported an inappropriately low glucocorticoid release in severe malaria [24]. The reason for this apparent HPA dysfunction in severe malaria is unclear, but could reflect low levels of the corticosteroid precursor cholesterol seen in malaria (see above), or a deregulation of the pituitary-adrenal function caused by *P. falciparum* parasites. Further research is needed to understand this phenomenon, but our results already suggest that

A. Metabolite signature of non-malaria febrile illness



Monte-Carlo cross-valid	lation in 200 subsamples						
avg. AUC	0.81 (95% CI: 0.80 - 0.82)						
avg. sensitivity	0.82 (95% CI:0.81 - 0.83)						
avg. specificity	0.79 (95% CI: 0.78 - 0.81)						
Permutation test (AUC): p - value = 0.01							

B. Metabolite signature of non-survival

ROC curve analysis on 15 cases and 45 controls



Metabolite test result for all 61 patients



Metabolite test result for all 61 patients





 $C_{21}H_{30}O_6$ 18-hydroxycortisol) and a C19-steroid-glucuronide (ID_3388_ON, m/z M-H = 481.243, putative ID: $C_{25}H_{38}O_9$ 11-beta-hydroxyandrosterone-3-glucuronide). Each panel shows: (i) a ROC curve along with the optimal cut-off value (the non-malaria analysis excluded BSI/malaria co-infection patients; the non-survival analysis excluded patients that left hospital against medical advice), (ii) check for over-fitting by Monte-Carlo cross-validation based on 200 rounds of balanced subsampling in all cases and controls and the significance permutation test, and (iii) a boxplot with the test result for all 61 patients (x-axis categories indicate the patient groups based on the study case definitions, y-axis shows the test-value of the diagnostic test which is based on the sum of the signal intensity of the two metabolites, here called composite signal intensity, included in each model, the orange dashed line indicates the cut-off value determined in the ROC analysis). (Abbreviations: LAMA = left against medical advice)

doi:10.1371/journal.pntd.0004480.g005

whatever the cause of the HPA dysfunction in malaria, it may be overcome by concomitant infections. Indeed, patients with the presumptive co-infections in our study were characterized by both the typical malaria lipids and the immunoregulatory metabolites found in non-malaria illness (Fig.3).

The immunoregulatory metabolites also appeared to be predictive of poor clinical outcome (non-survival) when present in high concentrations (Fig.3 and Fig.5B). There is indeed increasing evidence that an excessive proinflammatory response and an abnormal activation of the HPA axis are key determinants in progression to organ failure and death in patients with critical illness [25,26]. The high circulating levels of glucocorticoids are a symptom of either impaired glucocorticoid clearance or growing glucocorticoid tissue resistance which opens the door to uncontrolled systemic inflammation associated with high mortality [26].



Fig 6. Diagnostic performance of bile metabolite signature for BSI. The metabolite signature consists of 2 bile metabolites detected by C₁₈-UHPLC-MS: a C26 bile alcohol (ID_4741_ON, m/z M+FA-H = 483.331, putative ID: C26H46O5 27-Nor-5b-cholestane-3a,7a,12a,24,25-pentol bile alcohol) and a C27 bile acid (ID_6663_ON, m/z M+FA-H = 493.316, putative ID: C27H44O5 C27 bile acid). The figure includes: (i) a ROC curve along with the optimal cut-off value (excluding patients with BSI/malaria co-infection, incomplete or possible BSI diagnosis), (ii) check for over-fitting by Monte-Carlo cross-validation based on 200 rounds of balanced subsampling in all cases and controls and the significance permutation test, and (iii) a boxplot with the test result for all 61 patients (x-axis categories indicate the patient groups based on the study case definitions, y-axis shows the test-value of the diagnostic test which is based on the sum of the signal intensity of the two metabolites, here called composite signal intensity, included in each model, the orange dashed line indicates the cut-off value determined in the ROC analysis).

doi:10.1371/journal.pntd.0004480.g006

Diagnostic parameter	BSI tests												
	blood c	ulture *	16S sec	luencing	BSI sig	BSI signature							
	incl. all patients	incl. <5 yrs only	incl. all patients	incl. <5 yrs only	incl. all patients	incl. <5 yrs only							
Sensitivity (%)	52.6 (26.6-85.6)	47.3 (23.6–77.9)	71.9 (38.5–99.8)	67.8 (37.9–99.2)	81.4 (54.3–97.7)	98.1 (80.2–100)							
Specificity (%)	95.7 (80.1–100)	98.2 (84.1–100)	97.6 (81.0–100)	98.6 (84.7–100)	85.8 (62.8–99.9)	82.9 (54.7–99.9)							
PPV (%)	88.8 (46.9–99.9)	95.3 (58.2–100)	95.3 (54.6–100)	97.4 (68.6–100)	78.2 (37.0–99.9)	80.8 (42.2–99.9)							
NPV (%)	76.3 (48.6–95.1)	71.3 (45.4–91.8)	84.5 (52.5–99.9)	80.1 (52.3–99.7)	88.2 (62.7–98.7)	98.3 (81.0–100)							

Table 5. Estimation of diagnostic performance of BSI tests by Bayesian latent class models.

The analysis only included patients that had results of all 3 tests. Two analyses were performed (i) including all 51 patients, and (ii) including a subset of 38 patients < 5 years. Results are shown with 95% CI.

* Blood culture was considered the reference method of the test panel; abbreviations used: PPV = positive predictive value, NPV = negative predictive value.

doi:10.1371/journal.pntd.0004480.t005

BSI patients were marked by increased plasma levels of bile acids and alcohols, which appeared to be independent of the infecting bacterial species. Elevated plasma levels of bile acids points to endotoxemia-related cholestasis in the liver [27,28], and has also been reported as a marker of sepsis patients with community-acquired pneumonia [25]. Cholestasis and the associated elevated plasma levels of bile metabolites occurs in many conditions affecting liver function but in the target population of severe febrile illness it appeared to be fairly specific for BSI patients. With an estimated sensitivity and specificity > 80%, the BSI bile signature had (i) a superior sensitivity than the current diagnostic tests based on pathogen detection (sensitivity < 70%), (ii) a superior diagnostic performance than clinical assessment (estimated at sensitivity 83%, specificity 62%)[29], and (iii) a performance that very well approaches the reported minimal requirements that should be met by a BSI rapid diagnostic test to be costeffective in low-resource settings (sensitivity of 83%, specificity of 94%)[29]. The excellent sensitivity (98.1%) and negative predictive value (98.3%) in under-fives is particularly promising as those characteristics would allow the BSI test to be used as a screening test for patients with severe febrile illness attending primary health care service. The results of the test would allow first line health care workers to confidently withhold antibiotics from under-fives that are negative for the bile signature test, while those with a positive test result could be immediately referred to hospital for further diagnosis and care. Such a metabolite screening test needs to meet the 'ASSURED' criteria (affordable, sensitive, specific, user- friendly, rapid and robust, equipment free and delivered) to ensure that it can be used in resource-limited and remote settings [30]. Immunoassays generally meet ASSURED and are now available in two formats. The simple lateral-flow "dipstick" test is currently the most widely used format. One test-strip generally carries one or two antibodies for the detection of one or two biomarkers, and different tests can be combined in one device. A new emerging format is the microfluidic chip that allows measurement of multiple biomarkers. The chip is attached to mobile phones to read and display the test results to the healthcare worker. Prototypes of this digital test format are already being tested in Central Africa [31]. Metabolite detection with immunoassays has already been developed by the food & agriculture industry (e.g. dipstick tests to detect bacterial toxins, pesticides, veterinary drugs) whereby metabolite-binding antibodies are designed with computer-assisted molecular modelling [32].

The design of this study has several limitations. We acknowledge that the presented results are hypothesis-generating and large-scale validation studies are essential to validate the performance of the candidate diagnostic signatures for BSI and non-malaria illness. In this discovery

study, we did not include a diagnostic test panel for viral and non-malaria parasitic infections, which could have helped to better characterize the non-malaria patient group. We did not check the identified candidate diagnostic signatures in an asymptomatic control population and in patients with non-severe febrile illness. These 2 groups should be included in future validation studies. However a recent study conducted in Tanzania reported that over 70% of pediatric febrile patients without severe clinical signs had a viral disease [33], thus minimizing the urgency for diagnostic tests for this target population. Hence, we deem it a priority to validate the performance of the non-malaria and BSI signatures in a larger sample of patients with severe febrile illness.

In conclusion, this study demonstrates that malaria and non-malaria patients with severe febrile illness have some fundamental differences in host response that could be exploited for differential diagnosis of severe febrile illness. In combination with the current malaria RDT, a rapid test assessing the plasma levels of 3–4 metabolites could inform on the likelihood of malaria, non-malaria illness, BSI, and survival and could thus empower primary health care centers in making informed treatment and referral decisions.

Supporting Information

S1 Data. An excel sheet listing all the metabolite features (rows) characteristic for nonmalaria febrile illness, malaria, non-survival, and severe anemia. The excel sheet comprises the following information in the columns for each metabolite feature: (1) patient group (characterized by metabolite feature), (2) Feature ID (study ID of metabolite feature), (3) area under ROC curve (AUC), (4) sensitivity, (5) specificity, (6) sensitivity + specificity, (7) p-value Mann-Whitney test comparing cases and controls adjusted for multiple comparisons, (8) fold change (= ratio of median signal intensity cases and median signal intensity controls), (9) median signal intensity cases, (10) median signal intensity controls, (11) mass (determined by mass-spectrometry), (12) RT (chromatographic retention time in sec), (13) CoDa-DW (quality value of signal chromatogram determined by the CoDa algorithm which uses the Durbin-Watson statistic), (14) rsd clin. samp. (relative standard deviation of signal intensity in clinical samples), (15) rsd pbqc samples (relative standard deviation of signal intensity in pooled biological quality control samples), (16) formula (chemical formula of assigned putative identifications), (17) putative ID 1 (putative identification 1), (18) KEGG 1 (KEGG entry of putative identification 1), (19) HMDB 1 (HMDB entry of putative identification 1), (20) putative ID 2 (putative identification 2), (21) KEGG 2 (KEGG entry of putative identification 2), (22) HMDB 2 (HMDB entry of putative identification 2), (23) putative ID 3 (putative identification 3), (24) KEGG 3 (KEGG entry of putative identification 3), (25) HMDB 3 (HMDB entry of putative identification 3), (26) class (biochemical class of metabolite feature), (27) remarks on ID (additional information on identifications of metabolite feature), (28) remarks on signal (additional information on quality signal and related peaks = signals corresponding to derivative compounds), (29) MSI scoring (Metabolomics Standards Initiative score of identification with 1 = identified compound, 2 = putatively annotated compounds, 3 = putatively characterized compound classes, 4 = unknown compounds), (30) RT min (chromatographic retention time in min), (31) platform (platform with which the metabolite feature was detected). Metabolite features for which no plausible identification could be found are shaded in grey. Metabolites with a red Feature ID were upregulated in the considered patient group, metabolite features with a blue Feature ID were downregulated in the considered patient group. The putative identifications for the lipids were annotated with the abbreviations PC, PE, SL, MG, DG or TG for phosphatidylcholines, phosphatidylethanolamines, sphingolipids, monoacylglycerides, diacylglycerides and triacylglycerides respectively, and the total number of acyl side-chain carbons and the double

bonds in the side-chains. (XLS)

S1 Methods. Additional information on materials and methods including a detailed description of (i) sample processing, (ii) diagnostic procedures used in the study, (iii) metabolomics sample preparation and data acquisition, (iv) metabolomics data processing, and (v) statistical methods.

(DOCX)

S1 Results. Additional results describing the metabolite features characteristic for patients with severe anemia and in patients with poor clinical outcome (non-survival). (DOCX)

Acknowledgments

We thank the clinical staff of CMA and Nanoro Health District for their collaboration and for patient care, and the laboratory staff of CRUN for their contribution to the work.

Author Contributions

Conceived and designed the experiments: SDec JM SDeb JJ HT JRC. Performed the experiments: SDec SDeb NJWR RG TR MM. Analyzed the data: SDec JM SDeb NJWR. Contributed reagents/materials/analysis tools: SDeb NJWR MM RG HT JJ. Wrote the paper: SDec RG JJ JRC. Coordinated patient enrollment and diagnosis: JM GI KB MCT PL.

References

- Nadjm B, Amos B, Mtove G, Ostermann J, Chonya S, et al. (2010) WHO guidelines for antimicrobial treatment in children admitted to hospital in an area of intense Plasmodium falciparum transmission: prospective study. BMJ 340: c1350. doi: <u>10.1136/bmj.c1350</u> PMID: <u>20354024</u>
- Sigauque B, Roca A, Mandomando I, Morais L, Quinto L, et al. (2009) Community-acquired bacteremia among children admitted to a rural hospital in Mozambique. Pediatr Infect Dis J 28: 108–113. doi: <u>10.</u> <u>1097/INF.0b013e318187a87d</u> PMID: <u>19131902</u>
- Chheng K, Carter MJ, Emary K, Chanpheaktra N, Moore CE, et al. (2013) A prospective study of the causes of febrile illness requiring hospitalization in children in Cambodia. PLoS One 8: e60634. doi: <u>10.1371/journal.pone.0060634</u> PMID: <u>23593267</u>
- Punjabi NH, Taylor WR, Murphy GS, Purwaningsih S, Picarima H, et al. (2012) Etiology of acute, nonmalaria, febrile illnesses in Jayapura, northeastern Papua, Indonesia. Am J Trop Med Hyg 86: 46–51. doi: <u>10.4269/ajtmh.2012.10-0497</u> PMID: <u>22232450</u>
- Gething PW, Kirui VC, Alegana VA, Okiro EA, Noor AM, et al. (2010) Estimating the number of paediatric fevers associated with malaria infection presenting to Africa's public health sector in 2007. PLoS Med 7: e1000301. doi: <u>10.1371/journal.pmed.1000301</u> PMID: <u>20625548</u>
- Church J, Maitland K (2014) Invasive bacterial co-infection in African children with Plasmodium falciparum malaria: a systematic review. BMC Med 12: 31. doi: <u>10.1186/1741-7015-12-31</u> PMID: <u>24548672</u>
- Deen J, von Seidlein L, Andersen F, Elle N, White NJ, et al. (2012) Community-acquired bacterial bloodstream infections in developing countries in south and southeast Asia: a systematic review. Lancet Infect Dis 12: 480–487. doi: 10.1016/S1473-3099(12)70028-2 PMID: 22632186
- Reddy EA, Shaw AV, Crump JA (2010) Community-acquired bloodstream infections in Africa: a systematic review and meta-analysis. Lancet Infect Dis 10: 417–432. doi: <u>10.1016/S1473-3099(10)</u> <u>70072-4</u> PMID: <u>20510282</u>
- Maltha J, Guiraud I, Kabore B, Lompo P, Ley B, et al. (2014) Frequency of Severe Malaria and Invasive Bacterial Infections among Children Admitted to a Rural Hospital in Burkina Faso. PLoS One 9: e89103. doi: <u>10.1371/journal.pone.0089103</u> PMID: <u>24551225</u>
- Graham SM, English M (2009) Non-typhoidal salmonellae: a management challenge for children with community-acquired invasive disease in tropical African countries. Lancet 373: 267–269. doi: <u>10.1016/</u> <u>S0140-6736(09)60073-8</u> PMID: <u>19150705</u>

- Bottieau E, Gillet P, De Weggheleire A, Scheirlinck A, Stokx J, et al. (2013) Treatment practices in patients with suspected malaria in Provincial Hospital of Tete, Mozambique. Trans R Soc Trop Med Hyg 107: 176–182. doi: <u>10.1093/trstmh/trs012</u> PMID: <u>23382275</u>
- Phoba MF, De Boeck H, Ifeka BB, Dawili J, Lunguya O, et al. (2014) Epidemic increase in Salmonella bloodstream infection in children, Bwamanda, the Democratic Republic of Congo. Eur J Clin Microbiol Infect Dis 33: 79–87. doi: 10.1007/s10096-013-1931-8 PMID: 23975545
- Iqbal J, Siddique A, Jameel M, Hira PR (2004) Persistent histidine-rich protein 2, parasite lactate dehydrogenase, and panmalarial antigen reactivity after clearance of Plasmodium falciparum monoinfection. J Clin Microbiol 42: 4237–4241. PMID: 15365017
- Wongsrichanalai C, Barcus MJ, Muth S, Sutamihardja A, Wernsdorfer WH (2007) A review of malaria diagnostic tools: microscopy and rapid diagnostic test (RDT). Am J Trop Med Hyg 77: 119–127. PMID: 18165483
- Visser BJ, Wieten RW, Nagel IM, Grobusch MP (2013) Serum lipids and lipoproteins in malaria—a systematic review and meta-analysis. Malar J 12: 442. doi: <u>10.1186/1475-2875-12-442</u> PMID: <u>24314058</u>
- Surowiec I, Orikiiriza J, Karlsson E, Nelson M, Bonde M, et al. (2015) Metabolic Signature Profiling as a Diagnostic and Prognostic Tool in Pediatric Plasmodium falciparum Malaria. Open Forum Infect Dis 2: ofv062. doi: <u>10.1093/ofid/ofv062</u> PMID: <u>26110164</u>
- Parola P, Gazin P, Patella F, Badiaga S, Delmont J, et al. (2004) Hypertriglyceridemia as an indicator of the severity of falciparum malaria in returned travelers: a clinical retrospective study. Parasitol Res 92: 464–466. PMID: 14963768
- Webster JI, Sternberg EM (2004) Role of the hypothalamic-pituitary-adrenal axis, glucocorticoids and glucocorticoid receptors in toxic sequelae of exposure to bacterial and viral products. J Endocrinol 181: 207–221. PMID: <u>15128270</u>
- Silverman MN, Pearce BD, Biron CA, Miller AH (2005) Immune modulation of the hypothalamic-pituitary-adrenal (HPA) axis during viral infection. Viral Immunol 18: 41–78. PMID: <u>15802953</u>
- Ibrahim EA, Kheir MM, Elhardello OA, Almahi WA, Ali NI, et al. (2011) Cortisol and uncomplicated Plasmodium falciparum malaria in an area of unstable malaria transmission in eastern Sudan. Asian Pac J Trop Med 4: 146–147. doi: 10.1016/S1995-7645(11)60056-4 PMID: 21771440
- Shwe T, Khin M, Min H, Hla KK, Win YY, et al. (1998) Serum cortisol levels in patients with uncomplicated and cerebral malaria. Southeast Asian J Trop Med Public Health 29: 46–49. PMID: <u>9740267</u>
- Libonati RM, de Mendonca BB, Maues JA, Quaresma JA, de Souza JM (2006) Some aspects of the behavior of the hypothalamus-pituitary-adrenal axis in patients with uncomplicated Plasmodium falciparum malaria: Cortisol and dehydroepiandrosterone levels. Acta Trop 98: 270–276. PMID: 16846568
- Wilson M, Davis TM, Binh TQ, Long TT, Danh PT, et al. (2001) Pituitary-adrenal function in uncomplicated falciparum malaria. Southeast Asian J Trop Med Public Health 32: 689–695. PMID: 12041540
- Davis TM, Li TA, Tran QB, Robertson K, Dyer JR, et al. (1997) The hypothalamic-pituitary-adrenocortical axis in severe falciparum malaria: effects of cytokines. J Clin Endocrinol Metab 82: 3029–3033. PMID: 9284738
- Seymour CW, Yende S, Scott MJ, Pribis J, Mohney RP, et al. (2013) Metabolomics in pneumonia and sepsis: an analysis of the GenIMS cohort study. Intensive Care Med 39: 1423–1434. doi: <u>10.1007/</u> <u>s00134-013-2935-7</u> PMID: <u>23673400</u>
- 26. Annane D (2008) Adrenal insufficiency in sepsis. Curr Pharm Des 14: 1882–1886. PMID: 18691099
- Green RM, Beier D, Gollan JL (1996) Regulation of hepatocyte bile salt transporters by endotoxin and inflammatory cytokines in rodents. Gastroenterology 111: 193–198. PMID: <u>8698199</u>
- Trauner M, Arrese M, Lee H, Boyer JL, Karpen SJ (1998) Endotoxin downregulates rat hepatic ntcp gene expression via decreased activity of critical transcription factors. J Clin Invest 101: 2092–2100. PMID: <u>9593765</u>
- Penno EC, Crump JA, Baird SJ (2015) Performance Requirements to Achieve Cost-Effectiveness of Point-of-Care Tests for Sepsis Among Patients with Febrile Illness in Low-Resource Settings. Am J Trop Med Hyg.
- Peeling RW, Mabey D (2010) Point-of-care tests for diagnosing infections in the developing world. Clin Microbiol Infect 16: 1062–1069. doi: <u>10.1111/j.1469-0691.2010.03279.x</u> PMID: <u>20670288</u>
- Chin CD, Cheung YK, Laksanasopin T, Modena MM, Chin SY, et al. (2013) Mobile device for disease diagnosis and data tracking in resource-limited settings. Clin Chem 59: 629–640. doi: <u>10.1373/</u> <u>clinchem.2012.199596</u> PMID: <u>23327782</u>
- Xu ZL, Shen YD, Beier RC, Yang JY, Lei HT, et al. (2009) Application of computer-assisted molecular modeling for immunoassay of low molecular weight food contaminants: A review. Anal Chim Acta 647: 125–136. doi: <u>10.1016/j.aca.2009.06.003</u> PMID: <u>19591697</u>

33. D'Acremont V, Kilowoko M, Kyungu E, Philipina S, Sangu W, et al. (2014) Beyond malaria—causes of fever in outpatient Tanzanian children. N Engl J Med 370: 809–817. doi: <u>10.1056/NEJMoa1214482</u> PMID: <u>24571753</u>

University Library



A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s:

Decuypere, S; Maltha, J; Deborggraeve, S; Rattray, NJW; Issa, G; Berenger, K; Lompo, P; Tahita, MC; Ruspasinghe, T; McConville, M; Goodacre, R; Tinto, H; Jacobs, J; Carapetis, JR

Title:

Towards Improving Point-of-Care Diagnosis of Non-malaria Febrile Illness: A Metabolomics Approach

Date:

2016-03-01

Citation:

Decuypere, S., Maltha, J., Deborggraeve, S., Rattray, N. J. W., Issa, G., Berenger, K., Lompo, P., Tahita, M. C., Ruspasinghe, T., McConville, M., Goodacre, R., Tinto, H., Jacobs, J. & Carapetis, J. R. (2016). Towards Improving Point-of-Care Diagnosis of Non-malaria Febrile Illness: A Metabolomics Approach. PLOS NEGLECTED TROPICAL DISEASES, 10 (3), https://doi.org/10.1371/journal.pntd.0004480.

Persistent Link: http://hdl.handle.net/11343/262167

File Description: Published version License: CC BY